Author's response to reviews

Title: Causes and 3-year-incidence of blindness in Jing-An District, Shanghai, China 2001-2009.

Authors:

liangcheng wu (liangchengwu@yahoo.com.cn)
oxinghui sun (xhsun1962@yahoo.com.cn)
xingtao zhou (jitong0512@163.com)
chenghai weng (class1_1@163.com)

Version: 7 Date: 11 April 2011

Author's response to reviews:

Dear editor:
Thank you for your suggestion, we have revised the manuscripts point by point
Title: please ensure that the title is identical in the manuscript and the submission system
answer: we have revised the title in the submission system.

References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example `text [1].'
answer: we have revised it.

Tables: we notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references.
answer: we have included the tables in a tables section following the references.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Email address: please just use author initials when listing email addresses on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com
answer: yes we have revised

Highlighting/tracing: Please remove all highlighting and tracking from the manuscript
Yes # we have revised it

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements

yes # we have revised it.

Thanks
Liangcheng wu
Xinghuai sun